Author’s response to reviews

Title: Factors associated with myopia in Korean children: Korea National Health and Nutrition Examination Survey 2016–2017 (KNHANES VII)

Authors:

Hyuna Kim (landyonel@gmail.com)
Jin Seo (jinseok90@naver.com)
Woong-Sun Yoo (oocee@daum.net)
Gyu-Nam Kim (sayuh@naver.com)
Rock Kim (krb747@gmail.com)
Jae Chae (chaejaeeun@gmail.com)
Inyoung Chung (inyoung@gnu.ac.kr)
Seong-Wook Seo (stramast@naver.com)
Seong-Jae Kim (oocee@naver.com)

Version: 1 Date: 18 Jun 2019

Author’s response to reviews:

Dear editor,

The KNHANES is a nationwide, population-based, and cross-sectional health examination and survey conducted regularly since 1998, in Korea. Like previous survey (before 2015), the KNHANES 7 was reviewed by the Institutional Review Board of the Korea Centers for Disease Control and Prevention, the informed consent was obtained each participants including minors (under age 16), from the parents/guardians. I added this information in the ethical approval and consent to participate” section.